
Daily report for Oct 12, 2013
Log notes actually made at 8 am LST, but report entry was delayed by work being done on network back in Boulder.

Weather today was clear and sunny, with light winds.

Generator was filled by Tim on 10/11/2013. Should be checked/refilled in a few days.

Visual inspection of the site showed nothing of concern. Rass speakers are all working, as is the fan on the tower.

Power amp was as .148 mW

Profiler display looked good, with spectral profiles were all in alignment.

Nagios flags indicated critical warnings for the following, indicating that the samples were late:

Nidas_1_10

Nidas_1_11

WXT/standard/nc

Additionally, most of the lidar flags were still flagged as critical.

Phil Chilson arrived last night to relieve me. Clayton is scheduled to arrive today to relieve Tim.  Phil went to the 449 profiler site this afternoon and was 
working on analysis code and checking on the past performance of the the radar. Tim will be training both Phil and Clayton tomorrow on site operations. 
There is possibility of another IOP tomorrow evening.


	Daily report for Oct 12, 2013

